Author’s response to reviews

Title: Tuberculosis case finding based on symptom screening among immigrants, refugees and asylum seekers in Rome

Authors:

Monica Sañé Schepisi (monica.saneschepisi@inmi.it)
Gina Gualano (gualano@inmi.it)
Claudia Fellus (CLAUDIAFELLUS@ALICE.IT)
Nazario Bevilacqua (bevilacqua@inmi.it)
Marco Vecchi (mv7110@gmail.com)
Pierluca Piselli (pierluca.piselli@inmi.it)
Giuliana Battagin (batgini@yahoo.com)
Giulia Silvestrini (silvestrinigiulia@gmail.com)
Andrea Attanasio (andreattanasio@hotmail.com)
Alberto Vela (A.Vela@aslromah.it)
Giorgia Rocca (giorgia_rocca@hotmail.com)
Alessandro Rinaldi (a.rinaldi28@virgilio.it)
Pietro Benedetti (pbenedetti74@gmail.com)
Salvatore Geraci (s.geraci@areasanitaria.it)
Francesco N Lauria (lauria@inmi.it)
Enrico Girardi (enrico.girardi@inmi.it)

Version: 2 Date: 1 February 2013

Author’s response to reviews:

Editor
BMC Public Health
Rome, February 1st, 2013

Dear Editor,

Please find enclosed our manuscript entitled: “Tuberculosis case finding based on symptom screening among immigrants, refugees and asylum seekers in Rome” which we submit to your attention for publication as Original Article in the BMC Public Health.

As requested we are providing the English language version the questionnaire developed specifically for use in this study, and uploading it as additional file.

We are also uploading a revised version of the manuscript correctly formatted (page numbering and unjustified text) in journal style.

Correspondence should be addressed to:

Dr. Enrico Girardi